Author's response to reviews

Title: A 63-year-old patient with Chronic Lymphocytic Inflammation with Pontine Perivascular Enhancement Responsive to Steroids. Need for prolonged immunosuppressive therapy.

Authors:

Juerd Wijntjes (j.wijntjes@slaz.nl)
Ernest J. Wouda (e.wouda@slaz.nl)
Carl E.H. Siegert (c.siegert@slaz.nl)
Giorgos B. Karas (g.karas@slaz.nl)
Annemarie M.M. Vlaar (a.vlaar@slaz.nl)

Version: 2 Date: 7 November 2012

Author's response to reviews:

Revisions in response to reviewers Jay-jiguang Zhu and Wolfgang Grisold

- Unfortunately there were no representative T1/T1-contrast MRI images of our patient to add to the report. The same applies for axial images of the pons.
- We adjusted our neurological exam as suggested, also we deleted unnecessary details where possible.
- A native English speaker has read our report and made corrections where necessary.
- The title (after long consideration) remained unchanged. If an alternative title is Necessary, we propose the following, "Need for prolonged immunosuppressive therapy in CLIPPERS --- a case report".
- We added the complete cytology report of the cerebro spinal fluid.
- We added a table with lab results of the blood tests on admission and after the treatment with methotrexate and azathioprine as suggested.
- We completed the references.